

Is Gold Different for Islamic and Conventional Portfolios? A Sectorial Analysis

El khamlichi, Abdelbari and HOANG, Thi Hong Van and Wong, Wing-Keung

LERSEM, ENCG El Jadida, Chouaib Doukkali University, Morocco, Montpellier Business School, Montpellier Research in Management, France, Department of Finance and Big Data Research Center, Asia University, Taiwan, School of Mathematics and Statistics, North East Normal University, China

17 January 2017

Online at https://mpra.ub.uni-muenchen.de/76282/ MPRA Paper No. 76282, posted 17 Jan 2017 14:26 UTC

Is Gold Different for Islamic and Conventional Portfolios? A Sectorial Analysis

Abdelbari El khamlichi,^a Thi-Hong-Van Hoang,^{b*} Wing-Keung Wong,^c Zhenzhen Zhu^d

^a LERSEM, ENCG El Jadida, Chouaib Doukkali University, Morocco ^b Montpellier Business School, Montpellier Research in Management, France ^c Department of Finance and Big Data Research Center, Asia University, Taiwan ^d School of Mathematics and Statistics, North East Normal University, China

January 2017

Abstract

This article investigates the impact of gold in portfolios in distinguishing between Islamic and conventional stocks as well as between risk-averse and risk-seeking investors, while considering sectorial specificities. Using daily data from the Dow Jones indexes and the London gold market over the 2002-2014 period, the results obtained show that the stochastic dominance method is more robust than the mean-risk method to detect the difference between Islamic and conventional portfolios. For most sectors, risk-averters prefer conventional portfolios, while risk-seekers prefer Islamic portfolios. On the other hand, risk-averters prefer portfolios with gold, while risk-seekers prefer portfolios without gold. A robustness check on different sub-periods shows that these results are time-varying following the behavior of gold prices. These findings can provide useful information to investors respecting Sharia and looking for a diversification with commodities such as gold.

JEL Classifications: G11, C58

Keywords: Islamic vs. Conventional stocks; Risk-averse vs. Risk-seeking investors; Gold; Portfolio diversification; Mean-risk; Stochastic dominance.

Acknowledgments: We would like to thank participants at the 14th Infiniti Conference on International Finance (Trinity College Dublin, Ireland, June 13-14, 2016), and especially Professors Brian M. Lucey (Trinity College Dublin) and Maurice Peat (University of Sydney), for their valuable comments. We thank the members of the Finance group of Montpellier Research in Management for their comments on the first draft. We also thank Sally Burke and Chuck Haggerty for their proofreading. Any error and shortcoming remain the authors' responsibility.

^{*} *Corresponding author*: 2300, avenue des Moulins, 34185 Montpellier, France. Phone: +33 (0)4 67 10 28 02 - Fax: +33 (0)4 67 10 26 82.

Email addresses: A. El khamlichi (<u>abdelbari.el.khamlichi@gmail.com</u>), T.H.V. Hoang (<u>thv.hoang@montpellierbs.com</u>), W.K. Wong (<u>alanwkwong@gmail.com</u>), and Z.Z. Zhu (<u>zzzhu26@gmail.com</u>)

Introduction

The use of gold to diversify portfolios has been widely investigated during the last several decades, especially starting in 2010 following the studies of Baur and Lucey (2010) and Baur and McDermott (2010). Indeed, these studies demonstrate that gold can be a safe haven asset in stock portfolios. Following this idea, many other articles have investigated this role of gold in different countries and periods, e.g., Baur (2011), Ciner et al. (2013), Sadorsy (2014), Beckmann et al. (2015), Nguyen et al. (2016).¹ Overall, these studies show that gold acts as a safe haven for stocks and bonds. However, it is time-varying and market-specific. To the best of our knowledge, there has been no research studying whether gold is different for Islamic and conventional stocks. Though Nagayev et al. (2016) investigate the dynamic links between Islamic stocks and commodities (including gold), they have not considered the comparison with conventional stocks. Sadorsy (2014) distinguishes between socially responsible and conventional stocks, Islamic stocks have not been considered. Furthermore, there has been no distinction between risk-averse and risk-seeking investors in most of the previous studies on gold investments. In our opinion, this distinction is important because Hoang et al. (2015b) show that, in China, investments in gold are more suitable to risk-seeking investors than riskaverse ones.

As for Islamic finance, the first quantitative studies focus on Islamic mutual funds in Malaysia (Ismail and Shakrani, 2003). Their study has been followed by numerous studies comparing the performance of Islamic stocks to that of their conventional counterparts, after Dow Jones and the FTSE built the first Islamic indexes. The results of these studies are not unanimous. Some find that Islamic stocks outperform conventional ones (e.g., Arouri et al., 2013), while some others find evidence to the contrary (e.g., Al-Khazali et al., 2014). Furthermore, the majority of studies find that there is no significant difference between Islamic and conventional stocks (e.g., Dewandaru et al., 2015). To the best of our knowledge, there has been no study investigating the relationship between Islamic stocks and gold taking into consideration the distinction between sectors or between risk-averse and risk-seeking investors. For example, Ghazali et al. (2015) focus on the official gold and gold accounts compatible with Sharia (Islamic law) principles in Malaysia. However, they do not compare conventional and Islamic stocks. Furthermore, Abdullah et al. (2016) find that gold has a significant relationship with Islamic stocks in south-east Asian countries. However, they do not make a comparison with conventional stocks. In our opinion, this comparison should raise

¹ For a more complete literature review on the financial economics of gold, please refer to O'Connor et al. (2015).

interesting results because when considering the principles of Islamic finance (e.g., *riba*, *gharar*, *takaful*, etc.), one may think that there is a significant difference between Islamic stocks and conventional stocks in their relationship with gold (see Section 2 for more details about Islamic finance). Gold is considered in this comparison because it is an important commodity for Islamic finance, as shown by Lee (2011) and Mohammad Daud Bakar² at the Global Islamic Finance Forum in Kuala Lumpur (May 10-12, 2016).

In order to investigate the difference between Islamic and conventional stocks regarding their relationship with gold, our methodology is based on two different approaches: mean-risk (MR) (mean-variance and mean-MVaR)³ and stochastic dominance (SD), in distinguishing the preferences of risk-averse and risk-seeking investors. Using these two methods, we compare 4 types of portfolios: PF1 (100% stocks); PF2 (50% stocks+50% gold); PF3 (Markowitz minimal-variance portfolios); and PF4 (CCC-GARCH portfolios). Looking at these four portfolios, we carry out a double comparison: First, between Islamic and conventional portfolios; Second, between portfolios with gold and those without gold. In each case, the sectorial effect is taken into account. The daily data under study range from 31/12/2002 to 31/12/2014 with 2,942 observations. Data for stocks are taken from Dow Jones and data for gold prices are taken from London. To check the robustness of the results, we also test for the effect of time (based on 3 sub-periods).

The MR analysis shows that there is no significant difference between Islamic and conventional portfolios for most sectors (except for Financials and Utilities). The SD analysis shows that in most sectors, risk-averters prefer investing in portfolios with gold, while risk-seekers prefer those without gold, for both Islamic and conventional portfolios. Furthermore, risk-averse investors prefer conventional portfolios, while risk-seeking investors prefer the corresponding Islamic portfolios, in most sectors. The diversification strategies with gold can significantly change the relationships between Islamic and conventional portfolios. In addition, these results are time-dependent. Last, we find that gold is a safe haven asset for risk-averters in the period when gold prices were on an upward trend (2002-2011). However, it is not the case in a downward period (2011-2014) when both risk-averse and risk-seeking investors prefer to invest in portfolios without gold. These findings can provide useful information to investors respecting Sharia as well as investors looking for a better allocation between Islamic or conventional stocks and gold.

² Founder and chairman of <u>Amanie Advisors</u>.

 $^{^{3}}$ MVaR = Modified Value-at-Risk of Gregoriou and Gueyie (2003). These criteria for risk-averse and risk-seeking investors are also applied in Hoang et al. (2015b).

The rest of the paper is organized as follows. Section 2 presents a twofold literature review on gold investments and Islamic finance. Section 3 presents the data set while Section 4 details the methodology. The empirical results are discussed in Section 5, while Section 6 is devoted to a robustness check. Section 7 concludes.

2. Literature review: Gold investments and Islamic finance

This section aims to present a literature review on two different topics. The first one concerns the relationship between gold and stocks, and the second one concerns a comparison between Islamic and conventional stocks.

2.1. Gold in the diversification of portfolios

The first study to investigate gold investments was McDonald and Solnik (1977), several years after the abolition of the Bretton-Woods system. It was followed by Sherman (1982), Jaffe (1989), Chua et al. (1990), Blose (1996), Blose and Shieh (1995), Davidson et al. (2003), and Lucey et al. (2006). These studies reveal the significant relationship between gold and stocks, and the positive role of gold in the diversification of portfolios. Baur and Lucey (2010) and Baur and McDermott (2010) investigate the role of gold as a safe haven asset, which is defined as "*an asset that is negatively correlated (or uncorrelated) with another asset or portfolio in certain periods only, e.g., in times of falling stock markets.*" Following these two studies, many others, for example, Hood and Malik (2013), Beckmann et al. (2015), and Ghazali et al. (2015), Nguyen et al. (2016), examine the role of gold as a safe haven in different countries and periods.

Baur (2011) uses US data from 1979 to 2011 to conclude that gold evolved as a safe haven only recently. In assessing the impact of the Asian and global financial crises on precious metals over the period 1995-2010, Morales et al. (2011) show that precious metals are not affected by crises, except gold, which tends to generate effects on other precious metals. Creti et al. (2013) confirm that gold is a safe haven for stocks. Ciner et al. (2013) show that stocks, bonds, gold, and oil in the US and UK can be used as a safe haven for each other. Hood and Malik (2013) show that, unlike other precious metals, gold can serve as a hedge and a weak safe haven for the US stock market. Soucek (2013) finds that in unstable periods, the correlation between gold and equity tends to be weak or negative. Gold can thus serve as a safe haven as well as offer the benefit of diversification. However, Beckmann et al. (2015) find that the role of gold as a hedge and safe haven may be market-specific. Sadorsy (2014) reveals that gold and oil can also be used as a hedge and safe haven for socially responsible stocks, in a way similar to that for conventional stocks. Comparing gold to bonds, Flavin et al. (2014) find that both gold and longer-dated bonds can be considered as safe-haven assets. Applying the wavelet approach on daily data from 1980 to 2013, Bredin et al. (2015) conclude that gold acts as a safe haven for stocks and bonds only for horizons up to one year, but this was not true in the early 1980s. Chkili (2016) finds that gold is a safe haven for stocks in BRICS countries. Overall, the above-mentioned studies show that gold acts as a safe haven for stocks and bonds. However, it is time-varying and market-specific. These two findings are confirmed by Ghazali et al. (2015) and Nguyen et al. (2016). Ghazali et al. (2015) shows that domestic gold in Malaysia, in particular the Islamic gold account,⁴ is not a safe haven during episodes of extreme drops in the stock market.

Other studies do not investigate the specific role of gold as a safe haven but rather its impact on the diversification of portfolios. For example, Hammoudeh et al. (2013) find a significant relationship between gold and stocks. Gold can thus play an important role in the diversification of stock portfolios. Kumar (2014) shows that stock+gold portfolios perform better than stock-only portfolios. Based on a wavelet analysis, Michis (2014) concludes that gold provides the lowest contribution to the portfolios' risk at medium- and long-term investment horizons. Baur and Löffler (2015), Choudhry et al. (2015), and Malliaris and Malliaris (2015) confirm the results of previous studies about the significant impact of gold in the diversification of portfolios. Hoang et al. (2015a) find that portfolios with gold quoted on the Paris Stock Exchange stochastically dominate those without gold from 1949 to 2012. Hoang et al. (2015b) show that gold is more suitable to risk-seeking investors and in crisis periods in China, using data from the Shanghai Gold Exchange. Other papers confirm the significant relationship between gold and stocks, e.g., Narayan et al. (2013), Narayan et al. (2015), Raza et al. (2016), Jain and Biswal (2016), Kaabia et al. (2016), Basher and Sadorsy (2016), Zhang (2016), Barunik et al. (2016).

The contributions of our article to this literature concern our simultaneous distinction between Islamic and conventional stocks, between portfolios with and without gold, between risk-averse and risk-seeking investors, and also between 10 different sectors. Although Hoang et al. (2015b) also separate risk-averse and risk-seeking investors, they do not distinguish between Islamic and conventional stocks and between different sectors.

⁴ The Islamic gold account needs to meet the following conditions: (1) the exchange of one monetary form to another on a spot basis, (2) gold bought by investors will be deposited into a gold account, with the bank as the debtor and the investor as the creditor. Please refer to Ghazali et al. (2015), p. 194, for more details.

2.2. Islamic vs. conventional stocks

The literature on Islamic finance covers different aspects related to the application of Sharia and its impacts, for example in the banking system (Adebifar et al., 2013 and 2016; Beck et al., 2013), in financial markets (Kenourgios et al., 2016), in corporate governance (Mollah and Zaman, 2015), or in mutual funds (Abdelsalam et al., 2014). In this section, we will present first the landscape of Islamic finance before reviewing previous studies about the comparison between Islamic and conventional stocks.

A. Islamic financial landscape

The second angle of our study concerns the literature on Islamic finance. Indeed, ethical finance could be either morally or socially responsible. Islamic finance is faith-based and therefore considered morally responsible (Ghoul and Karam, 2007). The origin of Islamic finance can be traced back to the beginnings of Islam, but the development of modern Islamic financial institutions dates only from the middle of the 20th century. Hence, after some modest attempts to establish Islamic banks during the early sixties, the real beginning of the Islamic financial industry took place during the mid-seventies. In 1975, the Islamic countries. In the same year, the Dubai Islamic Bank was created to be the first commercial bank providing Sharia-compliant products and services to customers.

Islamic finance is based on a set of principles referred to as Sharia. These principles are the cornerstones of the theory and practice of Islamic finance. Hence, the prohibition of *riba* (illegitimate increase, or interest) is absolute, while agency contracts and partnerships are allowed. Islamic financial institutions reorient their activities on a non-interest basis, which means that profit- and loss-sharing (PLS) activities are encouraged. The allocation of financial resources on the basis of PLS gives a maximum weight to the investment profitability, whereas an interest-based allocation gives it to creditworthiness (Iqbal, 2002). Also, Islamic finance prohibits *gharar*, which means exposing oneself to excessive uncertainty in business activities (Iqbal, 2002). This excessive uncertainty could concern ambiguities in terms of the deal (regarding the price, quality, quantity, delivery date, etc.), exposing either of the parties to unnecessary risks. Another principle of Islamic finance is that each financial transaction must be tied to a tangible underlying asset (real estate or commodities, etc.). This is known as the asset-backing principle. In addition to the previous principles, Islamic financial institutions must avoid dealing with non-ethical sectors (such as alcohol, pornography, pork-related activities, tobacco, gambling, etc.).

In recent years, Islamic banking and finance has been recognized as a rapidly growing part of the financial sector. This is the result of the increasing attention to investments driven by innovations in Islamic finance. Actually, Islamic banks have received a lot of academic support and the literature on the subject is well documented. Hence, since the activities of Islamic banks are not based on interest, many researchers study to what extent these banks are different from conventional ones in terms of efficiency, risk, asset quality, and financial stability (e.g., Cihak and Hesse, 2010; Beck et al., 2013, Abedifar et al., 2013). Other researchers focus on their performance (e.g., Mollah and Zaman, 2015), as well as their contribution to financial development and economic welfare in Muslim countries (e.g., Abedifar et al., 2016). In addition to banking activities, Islamic finance covers many other segments, such as investment funds, money markets, stock markets, microfinance, the insurance industry (*takaful*), as well as Islamic indexes and securities.

B. Islamic mutual funds and indexes

The first wave of academic studies on Islamic financial markets takes a qualitative approach. They analyze the Islamic stock market in its early stages and its particularities in terms of practices and regulations (El gari, 1993; Naughton and Naughton, 2000; Obaidollah, 2001). With the development of data on Islamic financial stock markets, many papers start to adopt a quantitative approach either for mutual funds or indexes. As for Islamic mutual funds, the study of Ismail and Shakrani (2003) is among the first quantitative papers published in this field. Then, various studies are conducted in order to assess the performance of such ethical funds, but the benchmarks used are very different. Indeed, researchers use Islamic indexes (Muhammad and Mokhtar, 2008), conventional indexes (Mansor and Bhatti, 2011), or both indexes (Haddad et al., 2009). The results of these papers are very heterogeneous, since some of them find that Islamic mutual funds perform better during recession periods (Abdullah et al., 2007), while Hayat and Kraeussl (2011) document that Islamic funds significantly underperform during the last financial crisis. During expansion periods, Mansor and Bhatti (2011) find no difference, since they distinguish between two bullish periods and find that Islamic funds over-perform their benchmarks during the first period but underperform them during the second. In addition to the performance analysis, Abdelsalam et al. (2014) study the performance persistence of Islamic funds compared to their socially responsible counterparts. They find that the performance persistence exists for both types of funds, but only for the worst and best funds.

Since the inception of the Dow Jones and FTSE Islamic indexes, in February and October 1999, respectively, many researchers have conducted quantitative studies in this field. Most of the Islamic indexes are sub-indexes of global one.⁵ They are obtained using both qualitative (sector) and quantitative (financial) screens⁶ to filter out stocks and to assess whether they are compliant with Islamic principles (Sharia-compliant). As a result of the screening process, the asset universe gets relatively smaller for Sharia-compliant portfolio management (Derigs and Marzban, 2009). Furthermore, Islamic stocks are found to be less exposed to interest rate risk than their conventional counterparts.

The results of previous studies on the subject vary considerably. Some studies find that Islamic indexes and conventional ones are not significantly different from each other (Ahmad and Ibrahim, 2002; Abul-Hassan et al., 2005; Girard and Hassan, 2008; Dewandaru et al., 2015). In the meanwhile, others find that Islamic indexes generate higher returns than their conventional benchmarks (Walkshäusl and Lobe, 2012; Arouri et al., 2013; Mohammad and Ashraf, 2015, Charles and Darné, 2015). Narayan and Bannigidadmath (2015) find that meanvariance investors prefer investing in Islamic stocks. Furthermore, some studies (e.g., Ahmad and Ibrahim, 2002; Hussein and Omran, 2005) show that the performance level depends on market conditions (growth or decline periods). Regarding informational efficiency, some papers find that Islamic indexes are less efficient (Sensoy et al., 2015), while other studies find that they are at least as efficient as their conventional counterparts (Guyot, 2011; El khamlichi et al., 2014a). On the other hand, Islamic indexes are not considered homogeneous, since emerging Islamic stock markets seem to be less efficient than developed Islamic markets (Jawadi et al., 2015). Furthermore, neither the over-performance nor the underperformance of Islamic indexes, compared to their conventional counterparts, has gained scholars' unanimous support and the debates are not over yet (El khamlichi et al., 2014b).

The contribution of our paper to the current Islamic finance literature is threefold. First, to the best of our knowledge, there has been no study investigating the relationship between Islamic stocks and gold, taking into consideration sectorial distinctions. Though a recent paper (Ghazali et al., 2015) focuses on stocks and the official gold and gold accounts suitable to Islamic principles in Malaysia, neither Islamic stocks nor sectorial distinctions are investigated. Second, since Islamic screens could lead to a higher concentration in some

⁵ Except for the financial sector, for which all of the components are either Islamic banks or Islamic financial institutions.

⁶ More details regarding both screens are given in Section 3.

sectors (Charles and Darné, 2015), we also consider sectorial differences in our empirical analysis. For that, we use the ten industries included in the Industry Classification Benchmark (ICB) structure. Third, we use both the mean-risk and the stochastic dominance approaches in distinguishing between risk-averse and risk-seeking investors. Though Mensi et al. (2015) also use gold, the Dow Jones Islamic World Emerging Market index (DJIWEM), and Treasury bills (T-bills) to study their hedge and/or safe haven characteristics in 6 GCC stock markets, they do not investigate sectorial differences and those between risk-averse and risk-seeking investors.

The next section explains the data set used in our empirical analysis.

3. Data

Our data set is composed of daily values from December 31, 2002, to December 31, 2014. The stock sectors studied include Basic Materials, Consumer Goods, Consumer Services, Financials, Health Care, Industrials, Oil and Gas, Technology, Telecommunication, and Utilities. Islamic and conventional indexes for these sectors are formulated by Dow Jones and cover stocks worldwide. Gold prices are taken from the London gold market, the biggest spot gold market in the world. These prices are for one gold ounce (28.35 grams) quoted in the afternoon fixing since they include more information than the morning fixing (Hoang et al., 2016). All indexes and gold prices are nominal values expressed in USD. In total, there are 2,942 observations for each series.

Data for Islamic and conventional indexes are retrieved from the FactSet financial database. All Islamic indexes follow the same stock screening process using Sharia compliance criteria developed by Sharia scholars and regulators. The qualitative and quantitative screens are used to filter out stocks and to assess their compliance with Islamic principles. A qualitative screen is commonly known as a negative screen. It consists of excluding firms operating in specific activities. For the Dow Jones Islamic indexes, the screening is carried out using the Industry Classification Benchmark (ICB) for each firm. Therefore, this screening excludes non-ethical sectors such as alcohol, pornography, pork-related activities, tobacco, gambling, etc.

As for the quantitative screen, it is conducted using financial ratios. Indeed, after removing firms with unacceptable primary business activities, the remaining stocks are evaluated according to several financial ratio filters. The filters are based on criteria set up by the Dow Jones Islamic Market (DJIM) Index Sharia Supervisory Board to remove firms with unacceptable levels of debt or impure interest income. According to the guidelines of the DJIM indices (Dow Jones, 2015), all of the following must be less than 33%:

- 1. Total debt divided by trailing 24-month average market capitalization,
- 2. the sum of a firm's cash and interest-bearing securities divided by trailing 24month average market capitalization, and
- 3. accounts receivable divided by trailing 24-month average market capitalization.

Firms passing the qualitative and quantitative screens are included as components of the DJIM World Index and all its sub-indexes (regional, sectoral and capitalization-weighted indexes). Since the screening is very tough, it is interesting to study whether the performance of the Islamic indexes (obtained by using the Sharia compliance criteria) is different from that of the conventional indexes, as studied in this paper.

As an example, we plot indexes of the Financials and Utilities sectors as well as gold prices in Figure 1. To save space, the figures of the other sectors are presented in the Appendix file.





Note: For an easier comparison, the base of 100 is fixed at the beginning of the study period. The letter "I" indicates Islamic and "C" indicates "Conventional."

Gold prices reached their peak on September 6, 2011, at 1,895 USD per ounce. In general, Islamic indexes and their counterparts evolve very closely, each experiencing a tendency to increase between 2002 and 2008, a tendency to decrease in 2008, and again, a tendency to increase from 2009 to 2014. To take into account this time-varying characteristic, a robustness check will be performed to test whether the study period has an impact on the results. To do so, we divide the whole period into three different sub-periods:

- The first one, from December 31, 2002, to June 7, 2008, is characterized by the tendency of both stock and gold prices to increase;

- The second one, from June 8, 2008, to September 6, 2011, is characterized by the financial crisis period with a tendency for both stock and gold prices to decrease at the beginning of the crisis, followed by a tendency for gold prices to increase and peaked on September 6, 2011;

- The third one, from September 7, 2011, to December 31, 2014, is characterized by the tendency of stock prices to increase and gold prices to decrease.

For simplicity, we call sub-period 1 the "bull market for all assets," sub-period 2 the "bull market for gold," and sub-period 3, the "bear market for gold." Table 1 presents the basic descriptive statistics.

[Insert Table 1 here]⁷

Overall, the gold's rate of return is the highest (about 11% annually). Stock returns range between almost 3% and 9%, and the higher returns include stocks of the following sectors: Consumer Goods, Consumer Services, Health Care, Industrials, and Technology. The lowest ones concern the Financials and Telecommunication sectors. There is not a great gap between Islamic and conventional stock indexes. In most cases, the Islamic indexes' rates of return are lower than the conventional ones (for 6 over 10 sectors). The sectors for which Islamic stocks are more profitable are Consumer Services, Financials, Industrials, and Utilities. The standard deviation is the lowest for the Consumer Goods and Health Care sectors (about 13%). The highest ones are for Basic Materials and Oil & Gas (about 24%). The one for gold is close to 20% per year. Once again, we do not find a great gap between Islamic and conventional stocks, except for the Utilities and Financial sectors (21% vs. 15%, and 25% vs. 21%, respectively). For the Utilities sector, this gap can be explained by the fact that most of the firms in this sector are related to energy activities that depend on the oil market, which is, in turn, greatly related to Islamic countries (such as the Gulf countries). As for the Financials sector, this gap can be explained by the fact that the activities in this sector diverge the most between Islamic and conventional firms. As for the skewness coefficients, they are significantly negative, indicating that the distributions of returns are skewed to the left. The kurtosis excess coefficients are significantly positive, indicating that the distributions of returns are leptokurtic. All normality tests show that the distributions are not normal.

To summarize, basic descriptive statistics show that there is not a significant gap between Islamic stocks and conventional stocks, except for the Utilities and Financial sectors. In this case, would their relationship with gold be different? The next section will focus on the methodology that allows answering this question.

⁷ Please refer to the Appendix file for all tables of results.

4. Methodology

To study whether gold is different with Islamic and conventional stocks, we follow 4 steps:

- First, we constitute four portfolios: PF1 (100% stocks),⁸ PF2 (50% stocks and 50% gold), PF3 (Markowitz minimal-variance portfolios), and PF4 (CCC-GARCH portfolios). As we mentioned above, there is a double comparison: (1) between Islamic and conventional portfolios, and (2) between portfolios with gold and those without gold.

- Second, the double comparison is made through mean-risk (MR) criteria (mean-variance and mean-MVaR) in distinguishing between risk-averse and risk-seeking investors.

- Third, to circumvent the limits of the MR approach, the stochastic dominance (SD) approach is used in distinguishing between risk-averse and risk-seeking investors. Indeed, this method allows us to compare the entire distributions of returns and not only the first few moments of the distribution, such as the mean, variance, skewness and kurtosis. Moreover, Lean et al. (2008) show that the SD approach is robust to non-i.i.d. data, including heteroscedastic data, which is usually the case with financial data.

- Fourth, for a robustness check, we will conduct the above analysis in three sub-periods to test whether the results are time-dependent.

All of the above-mentioned methods are detailed in the following sub-sections. The first one is devoted to the optimal weight of gold in PF4 following the Kroner and Ng (1998) method. The second and third sub-sections focus on the mean-risk analysis (mean-variance and mean-MVaR) from which the optimal weight of gold in PF3 can be calculated. The fourth sub-section is dedicated to the stochastic dominance method.

4.1. Bivariate-GARCH models and the optimal weight

To determine the optimal weight of gold and stocks in PF4, we employ the method proposed by Kroner and Ng (1998) model⁹ as follows:

$$w_t^G = \frac{h_t^S - h_t^{SG}}{h_t^G - 2h_t^{SG} + h_t^S}$$

where w_t^G is the optimal weight of gold in the portfolio, h_t^P is the conditional variance of the stock-only portfolio *S*, h_t^{SG} is the conditional covariance between the stock-only portfolio and gold, and h_t^G is the conditional variance of gold. The optimal weight is thus calculated for each date under the condition that: $w_t^G = 0$ if $w_t^G < 0$, $w_t^G = w_t^G$ if $0 \le w_t^W \le 1$, and $w_t^G = 1$ if $w_t^G > 1$. The optimal weight used is thus the average value over our study period.

We base our study on the bivariate CCC-GARCH(1,1) model of Bollerslev (1990) to estimate the conditional variances and covariance. We choose the CCC representation because it provides more economic significance in estimating conditional correlation rather than the conditional covariance (as in the BEKK-GARCH model of Engle and Kroner (1995),

⁸ In the composition of these portfolios, stocks are referred to as the indexes (Islamic or conventional).

⁹ For another application of this method, please refer to Arouri et al. (2011).

for example). For each pair of returns of the stock-only portfolio and gold, the bivariate VAR(1)-GARCH(1,1) has the following specification:

$$\begin{cases} R_t = \mu + \Phi R_{t-1} + \varepsilon_t \\ \varepsilon_t = H_t^{1/2} \eta_t \end{cases}$$

where $R_t = (R_t^S, R_t^G)'$ is the vector of returns of the stock-only portfolio and gold, respectively, Φ refers to a (2 x 2) matrix of coefficients $\Phi = \begin{pmatrix} \phi_1 & 0 \\ 0 & \phi_2 \end{pmatrix}$, $\varepsilon_t = (\varepsilon_t^S, \varepsilon_t^G)$ is the vector of the error terms of the conditional mean equations for the stock-only portfolio and gold, respectively, $\eta_t = (\eta_t^S, \eta_t^G)$ refers to a sequence of independently and identically distributed (*i.i.d.*) random errors with $E(\eta_t) = 0$ and $Var(\eta_t) = I_N$, and $H_t = \begin{pmatrix} h_t^S & h_t^{SG} \\ h_t^{SG} & h_t^G \end{pmatrix}$ is the matrix of conditional variances of returns of the stock-only portfolio and gold.

The CCC-GARCH(1,1) model specifies the H₁ matrix as follows:

$$H_t = D_t K D_t$$

with $D_t = diag(\sqrt{h_t^S}, \sqrt{h_t^G})$, and $K = (\rho_{ij})$ is the (2 x 2) matrix containing the constant conditional correlations ρ_{ij} with $\rho_{ii} = 1$, i = S, W. The conditional variances and covariance are given by:

$$\begin{cases} h_t^{S} = C_S + \alpha_P (\varepsilon_{t-1}^{S})^2 + \beta_S h_{t-1}^{S} \\ h_t^{G} = C_G + \alpha_G (\varepsilon_{t-1}^{G})^2 + \beta_G h_{t-1}^{G} \\ h_t^{SG} = \rho \sqrt{h_t^{S} h_t^{G}} \end{cases}$$

To estimate this model, the maximum likelihood method is used.

4.2. Mean-variance analysis for risk-averse and risk-seeking investors

The classical mean-variance portfolio optimization (MVPO) model introduced by Markowitz (1952), and improved later by Bai et al. (2009), Leung et al. (2012), and others, can be used to determine the asset allocation for a given amount of capital through the efficient frontier. To present the MVPO model formally, we assume that there are *n* assets and let x_i (*i*=1,...,*n*) be the fraction of the capital invested in asset *i* of portfolio *P* in which the average return R_p is maximized, subject to a given level of its variance σ_p^2 . We denote R_i to be the expected return of asset *i* and σ_{ij} the covariance of returns between assets *i* and *j*, for any *i*, *j* =1,...,*n*. The general MVPO model is presented as follows:

Max
$$R_p = \sum_{i=1}^{n} R_i x_i$$
, subject to: $\sum_{i=1}^{n} \sum_{j=1}^{n} \sigma_{ij} x_i x_j = \sigma_p^2$ and $\sum_{i=1}^{n} x_i = 1$.

From this efficient frontier, we will use the minimal-variance portfolio (PF3) which the lowest point on the curve. As for the mean-variance (MV) comparison, we consider the returns of any two portfolios Y and Z, with means μ_y and μ_z and standard deviations σ_y and σ_z . Y is said to dominate Z by the MV rule for risk-averters (Markowitz, 1952) if $\mu_y \ge \mu_z$ and $\sigma_y \le \sigma_z$. Y is said to dominate Z by the MV rule for risk-seekers (Wong, 2007) if $\mu_y \ge \mu_z$ and $\sigma_y \ge \sigma_z$ and if the inequality holds in at least one of the two conditions.

4.3. Mean-MVaR analysis for risk-averse and risk-seeking investors

It is well-known that downside risk is a major factor in measuring financial risk. In the meanvariance (MV) context of Markowitz (1952), the variance or standard deviation is adopted to measure the risk exposure of financial assets. Nonetheless, it cannot capture the downside risk of an asset. To circumvent this limitation, Jorion (2000) proposes using the Value-at-Risk (VaR); Rockafellar and Uryasev (2000) recommend using the conditional-VaR (CVaR); while Gregoriou and Gueyie (2003) suggest using the modified VaR (MVaR). The limitation of the traditional VaR is the use of a symmetrical distribution function, which is not the case for our series (see Section 3). Thus, to compare the risk of different portfolios, we choose to use the MVaR (Gregoriou and Gueyie, 2003), which takes higher moments of the distribution of returns into account (skewness and kurtosis). Moreover, this measure is appropriate to our portfolios because of their weak skewness, as pointed out by Caporin et al. (2014).

The MVaR is defined by: $MVaR_{1-\alpha} = \mu - \sigma Z_{CF,\alpha}$ where $1-\alpha$ is the confidence level of the MVaR, μ and σ are the mean and standard deviation of the asset returns, and $Z_{CF,\alpha}$ is the Cornish-Fisher approximation of the α % quantile of the distribution:

$$Z_{CF,\alpha} = Z_{\alpha} + \frac{1}{6}S(Z_{\alpha}^{2} - 1) + \frac{1}{24}K(Z_{\alpha}^{3} - 3Z_{\alpha}) - \frac{1}{36}S^{2}(2Z_{\alpha}^{3} - 5Z_{\alpha})$$

where Z_{α} is the $\alpha\%$ quantile of a standard normal distribution, S is the skewness, and K is the excess kurtosis. In our paper, we use the conventional quantile $\alpha\% = 5\%$.

Ogryczak and Ruszczynski (2002) show that under the expected-utility principle, the VaR criterion is equivalent to the first-order SD (FSD) for risk-averters, while Ma and Wong (2010) prove that the CVaR criterion is equivalent to the second-order SD (SSD) for risk-averters. If Y dominates Z in the sense of the FSD and SSD, then it is well known that the mean of Y is higher than that of Z. On the other hand, Levy (2015) shows that under some conditions (for example, equal mean), the order of SSD for risk-seekers is reversed from that of the SSD for risk-averters. Chan et al. (2012a) find that in some situations, the second and third order SD for risk-seekers is reversed from that for risk-seekers is the same as that for risk-averters. In some other situations, the second and third order of the MVaR is consistent with that of the CVaR under some conditions, we make the following rules:

- 1) If $\mu_Y \ge \mu_Z$ and $MVaR(Y) \le MVaR(Z)$, then risk-averters prefer Y to Z,
- 2) If $\mu_{Y} \ge \mu_{Z}$ and $MVaR(Y) \ge MVaR(Z)$, then risk-seekers prefer Y to Z.

We call (1) the mean-MVaR rule for risk-averters and (2) the mean-MVaR rule for risk-seekers.

4.4. Stochastic dominance analysis for risk-averse and risk-seeking investors

The choice to include the SD approach is motivated by various reasons. First, it can be used to draw inferences on the expected utility maximization for different types of investors. It then allows investors to appropriately rank portfolios without any assumption on the distribution of returns, and only a few very simple assumptions on utility functions.¹⁰ Indeed, the SD method incorporates information on the entire distribution of returns, rather than only the first few moments. Second, the SD method requires no precise assessment as to the specific form of investors' risk preference or utility functions. Thus, in our study, we employ the SD approach to compare the cumulative distribution functions of returns for the portfolios under study. If one portfolio stochastically dominates another, then an investor would prefer the dominating portfolios to maximize his or her expected utility and/or wealth, following the order of the SD, as explained below (Sriboonchitta et al., 2009; Levy, 2015).

To apply the SD approach, we let Y and Z be the return series of two portfolios with a common support of [a, b] (a < b), cumulative distribution functions (CDFs), F and G, and corresponding probability density functions (PDFs), f and g, respectively, and we define:

¹⁰ The first and third derivatives of the utility function are positive, while the second one is negative for risk-averters and positive for risk-seekers.

$$H_{0} = H_{0}^{R} = h, \ H_{j}(x) = \int_{a}^{x} H_{j-1}(t) dt, \text{ and } H_{j}^{R}(x) = \int_{x}^{b} H_{j-1}^{R}(t) dt$$
(1)

for h = f, g; H = F, G; for any integer *j*. We call the integral H_j [H_j^R] the *j*th-order [reversed] integral for H = F and *G*. We note that the *j*th-order [reversed] integral defined in (1) corresponds to the *j*th-order [reversed] SD for risk-averters [-seekers], which is defined as follows: *Y* dominates *Z* by FSD (SSD, TSD) [FRSD (SRSD, TRSD)],¹¹ denoted by $Y \succ_1 Z$ ($Y \succ_2 Z$, $Y \succ_3 Z$) [$Y \succ_1^R Z$ ($Y \succ_2^R Z$, $Y \succ_3^R Z$)] if and only if $F_1(x) \le G_1(x)$ ($F_2(x) \le G_2(x), F_3(x) \le G_3(x)$) [$F_1^R(x) \ge G_1^R(x)$ ($F_2^R(x) \ge G_2^R(x), F_3^R(x) \ge G_3^R(x)$)] for all possible returns *x*, and the strict inequality holds for some values of *x*, where FSD (SSD, TSD) denotes first- (second-, third-) order traditional SD, and FRSD (SRSD, TRSD) denotes first- (second-, third-) order reversed SD, respectively. We note that Levy (2015) calls the SD theory for risk-seekers the "risk seeking SD theory" and denotes it as RSSD. In this paper, we follow Levy (2015) and call it risk-seeking SD, but we denote it as RSD (instead of RSSD).

For any integer *j*, SD corresponds to three broadly defined utility functions, U_j , for risk-averters, and RSDs correspond to three broadly defined utility functions, U_j^R , for risk-seekers. U_j is the set of utility functions such that $U_j = \{u : (-1)^{i+1}u^{(i)} \ge 0, i = 1, \dots, j\}$. U_j^R is the set of utility functions such that $U_j^R = \{u : u^{(i)} \ge 0, i = 1, \dots, j\}$, where $u^{(i)}$ is the ith derivative of the utility function *U*. It is well-known that for any integer *j*, $Y \succ_j Z$ ($Y \succ_3^R Z$) if and only if $E[u(X)] \ge E[u(Z)]$ for any *u* in $U_j(U_j^R)$ (Li and Wong, 1999; Guo and Wong, 2016). Thus, risk-averse investors exhibit FSD (SSD, TSD) if their utility functions *u* belong to U_1 (U_2, U_3). On the other hand, risk-seeking investors exhibit RFSD (RSSD, RTSD) if their utility functions *u* belong to $U_1^R(U_2^R, U_3^R)$.

In practice, the above definitions imply that risk-averters prefer portfolios that have a lower probability of loss (lower CDF or lower risk), whereas risk-seekers prefer portfolios that have a higher probability of gain (higher reversed CDF). In other words, to make a choice between two portfolios, risk-averters will choose the one with a lower CDF, since it has a lower probability of loss (e.g., $F_1(x) \le G_1(x)$). On the other hand, risk-seekers will choose the one with a higher reversed CDF, since it has a higher probability of gain (e.g., $F_1^R(x) \ge G_1^R(x)$). To date, SD tests have been well developed (e.g., Davidson and Duclos, or DD, 2000). Since the DD test was found to be powerful, less conservative in size, and robust to non-i.i.d. and heteroscedastic data (Lean et al., 2008), we employ the DD test and its extension as presented in the following subsections.

A. Stochastic dominance test for risk-averters (SD)

Assume the data $\{(f_i, g_i)\}$ for $i = 1, 2, \dots N_h$ are observations drawn from any two returns of portfolios *Y* and *Z* with CDFs *F* and *G*, respectively. For a grid of pre-selected points $x_1, x_2...x_k$, Bai et al. (2015) modify the statistic developed by Davidson and Duclos (2000) to obtain the following j^{th} -order SD test statistic, $T_i(x)$ (j = 1, 2, and 3), for risk-averters:

¹¹ In these definitions, F denotes First Order, S denotes Second Order, T denotes Third Order and R denotes Reversed. Readers may refer to Chan et al. (2012a) and Guo and Wong (2016) for more details on the definitions.

$$T_{j}(x) = \frac{\hat{F}_{j}(x) - \hat{G}_{j}(x)}{\sqrt{\hat{V}_{j}(x)}} , \qquad (2)$$

where

$$\begin{split} \hat{V}_{j}(x) &= \hat{V}_{F_{j}}(x) + \hat{V}_{G_{j}}(x) - 2\hat{V}_{FG_{j}}(x);\\ \hat{H}_{j}(x) &= \frac{1}{N_{h}(j-1)!} \sum_{i=1}^{N_{h}} (x-h_{i})_{+}^{j-1},\\ \hat{V}_{H_{j}}(x) &= \frac{1}{N_{h}} \left[\frac{1}{N_{h}((j-1)!)^{2}} \sum_{i=1}^{N_{h}} (x-h_{i})_{+}^{2(j-1)} - \hat{H}_{j}(x)^{2} \right], H = F, G; h = f, g;\\ \hat{V}_{FG_{j}}(x) &= \frac{1}{N_{h}} \left[\frac{1}{N_{h}((j-1)!)^{2}} \sum_{i=1}^{N_{h}} (x-f_{i})_{+}^{j-1} (x-g_{i})_{+}^{j-1} - \hat{F}_{j}(x)\hat{G}_{j}(x) \right], \end{split}$$

in which F_i and G_i are defined in (1).

It is empirically impossible to test the null hypothesis for the full support of the distributions. Thus, Bishop et al. (1992) propose testing the null hypothesis for a pre-designed finite number of values x. Specifically, for all i = 1, 2, ..., k, the following hypotheses are tested:

$$H_{0}: F_{j}(x_{i}) = G_{j}(x_{i}), \text{ for all } x_{i};$$

$$H_{A}: F_{j}(x_{i}) \neq G_{j}(x_{i}) \text{ for some } x_{i};$$

$$H_{A1}: F_{j}(x_{i}) \leq G_{j}(x_{i}) \text{ for all } x_{i}, F_{j}(x_{i}) < G_{j}(x_{i}) \text{ for some } x_{i};$$

$$H_{A2}: F_{i}(x_{i}) \geq G_{i}(x_{i}) \text{ for all } x_{i}, F_{j}(x_{i}) > G_{j}(x_{i}) \text{ for some } x_{i}.$$

We note that in the above hypotheses, H_A is set to be exclusive of both H_{A1} and H_{A2} , such that if either H_{A1} or H_{A2} is accepted, this does not mean that H_A is accepted. Bai et al. (2015) suggest using a simulation approach to generate the simulated critical values. In this paper, we follow their recommendation.

The SD test compares the distributions of Y and Z at a finite number of grid points, and various studies have examined the choice of these points. Too few grids will miss information on the distributions between any two consecutive grids (Barrett and Donald, 2003). To make more detailed comparisons, we follow Fong et al. (2008), and others, to make eleven equal-spaced major grids (10 partitions) and nine equal-spaced minor grids (10 partitions) within any two consecutive major partitions. We then have a total of 101 grids with 100 equal-spaced partitions in each comparison. Bai et al. (2015) improve the SD test by deriving the limiting process of the SD statistic $T_j(x)$ so that the SD test can be performed by $max_x |T_j(x)|$ to take care of the dependency of the partitions. We follow their recommendation in our analysis.

B. Stochastic dominance test for risk-seekers (RSD)

We follow Bai et al. (2015) to use the SD test statistic for risk-seekers and call it the riskseeking SD (or RSD) test statistic. Let $\{(f_i, g_i)\}$ for $i = 1, 2, \dots N_h$ be observations drawn from Y and Z. For a grid of pre-selected points $x_1, x_2... x_k$, Bai et al. (2015) modify the statistic developed by Davidson and Duclos (2000) to obtain the following j^{th} -order RSD statistics T_i^R for risk-seekers (*R* denotes risk-seeking behavior) such that:

$$T_{j}^{R}(x) = \frac{\hat{F}_{j}^{R}(x) - \hat{G}_{j}^{R}(x)}{\sqrt{\hat{V}_{j}^{R}(x)}}$$
(3)

where $\hat{V}_{j}^{R}(x) = \hat{V}_{F_{j}}^{R}(x) + \hat{V}_{G_{j}}^{R}(x) - 2\hat{V}_{FG_{j}}^{R}(x);$

$$\hat{H}_{j}^{R}(x) = \frac{1}{N_{h}(j-1)!} \sum_{i=1}^{N_{h}} (h_{i} - x)_{+}^{j-1},$$

$$\hat{V}_{H_{j}}^{R}(x) = \frac{1}{N_{h}} \left[\frac{1}{N_{h}((j-1)!)^{2}} \sum_{i=1}^{N_{h}} (h_{i} - x)_{+}^{2(j-1)} - \hat{H}_{j}^{R}(x)^{2} \right], H = F, G; h = f, g;$$

$$\hat{V}_{FG_{j}}^{R}(x) = \frac{1}{N_{h}} \left[\frac{1}{N_{h}((j-1)!)^{2}} \sum_{i=1}^{N_{h}} (f_{i} - x)_{+}^{j-1} (g_{i} - x)_{+}^{j-1} - \hat{F}_{j}^{R}(x) \hat{G}_{j}^{R}(x) \right];$$

and the integrals $F_j^R(x)$ and $G_j^R(x)$ are defined in (1) for j = 1, 2, 3. For i = 1, 2, ..., k, the following hypotheses are tested for risk seekers:

$$\begin{split} H_0: F_j^R(x_i) &= G_j^R(x_i) \text{, for all } x_i; \\ H_R: F_j^R(x_i) &\neq G_j^R(x_i) \text{ for some } x_i \text{ ;} \\ H_{R1}: F_j^R\left(x_i\right) &\geq G_j^R\left(x_i\right) \text{ for all } x_i, F_j^R\left(x_i\right) > G_j^R\left(x_i\right) \text{ for some } x_i; \\ H_{R2}: F_j^R\left(x_i\right) &\leq G_j^R\left(x_i\right) \text{ for all } x_i, F_j^R\left(x_i\right) < G_j^R\left(x_i\right) \text{ for some } x_i. \end{split}$$

We construct grid points for the RSD test (for risk-seekers) in the same way as for the SD test (for risk-averters) to get 101 grids with 100 equal-spaced partitions in each comparison. We also follow the approach recommended by Bai et al. (2015) to obtain the simulated critical values for the RSD test.

Not rejecting either H_0 or H_A or H_R implies: (1) there is no SD between Y and Z; (2) there is no arbitrage opportunity between these two portfolios, and thus, no portfolio is preferred to the other. If H_{A1} (H_{A2}) [H_{R1} (H_{R2})] of order one is accepted, Y (Z) stochastically dominates Z (Y) at the first order for risk-averters [-seekers]. In this situation, and under certain regularity conditions,¹² an arbitrage opportunity exists and investors will be better off switching from the dominated to the dominant portfolio. These results imply that neither the market efficiency nor the market rationality holds in these markets (Chan et al., 2012b; Qiao et al., 2013). On the other hand, if H_{A1} (H_{A2}) [H_{R1} (H_{R2})] is accepted at order two (three), a particular portfolio stochastically dominates the other at the second (third) order. In this situation, an arbitrage opportunity does not exist, and switching from one portfolio to another will only increase the expected utility of risk-averters [-seekers], but not their expected wealth (Jarrow, 1986; Falk and Levy, 1989; Wong et al., 2008). These results imply that both market efficiency and market rationality could still hold in these markets.

In order to minimize Type II errors and to accommodate the effect of almost SD,¹³ we follow Gasbarro et al. (2007), among others, and use a conservative 5% cut-off point in checking the proportion of test statistics for statistical inference. Using a 5% cut-off point implies that one prospect dominates another only if at least 5% of the statistics are significant.

¹² Please refer to Jarrow (1986) for the conditions.

¹³ Readers may refer to Leshno and Levy (2002) and Guo et al. (2014, 2016) and the references therein for more information. Leshno and Levy (2002) use an example of 1% violation to state the problem of almost SD.

5. Results and discussions

As mentioned in Section 4, we build 4 types of portfolios: PF1 (100% stocks), PF2 (50% stocks+50% gold), PF3 (Markowitz minimal-variance portfolios), and PF4 (CCC-GARCH portfolios). In this section, we will first present the results on the weights of gold in PF3 and in PF4. Once all the portfolios are built, we will then conduct the double comparison between Islamic portfolios and conventional portfolios (for both with and without gold) and between portfolios with and without gold (for both Islamic and conventional portfolios). The MR results are presented first, followed by SD results.

5.1. The weight of gold in portfolios PF3 and PF4

As explained in Section 4.1, PF3 is the minimal-variance portfolio (Markowitz, 1952) for each sectorial stock portfolio (either Islamic or conventional) diversified by gold. The weights of gold in these portfolios are presented in Table 2.

[Insert Table 2 here]

As mentioned in Section 4.1, we use the CCC-GARCH model to calculate the optimal weight of gold in PF4 (either Islamic or conventional). Table 3 presents the results.

[Insert Table 3 here]

In both PF3 and PF4, the weight of gold is higher in Islamic portfolios than in conventional ones. However, this difference is not very high (between 1% and 5%). Nevertheless, for the Utilities sector, there is a very high difference: 53% for Islamic and 33% for conventional portfolios in PF3, while the numbers are 50% and 33% in PF4, respectively. These high weights of gold in portfolios minimizing the risk (measured by the variance in PF3 and conditional variance in PF4) confirm the results of previous studies about the positive role of gold in the diversification of portfolios (e.g., Choudhry et al., 2015; Nguyen et al., 2016).

As we now have the composition of the four portfolios, the following sections will focus on the double comparison between Islamic and conventional portfolios and between portfolios with gold and those without gold

5.2. Mean-risk comparisons

5.2.1. Mean-risk comparisons between Islamic and conventional portfolios

The tables below present the results on mean-variance and mean-MVaR analyses to compare the performance of both Islamic and conventional portfolios for the four kinds of portfolios (PF1, PF2, PF3, and PF4) defined in Section 4. Table 4 presents results on PF1, Table 5 on PF2, Table 6 on PF3 and Table 7 on PF4.

[Insert Tables 4, 5, 6, and 7 here]

From Table 4 (PF1, 100% stocks), the MV results show that the average returns are not significantly different in all sectors. However, for half of the sectors (Basic Materials, Consumer Services, Financials, Telecommunications, and Utilities), the variances of Islamic stocks are significantly greater than those of the conventional ones. For the other half, the variances are not significantly different. The two highest differences are for the Utilities and Financial sectors (20% vs. 14% and 25% vs. 21%, respectively, with a p-value less than 1%). The conclusion drawn from the mean-MVaR is similar to that from the mean-variance analysis. Table 5 (PF2, 50% stocks+50% gold) shows that there is no significant difference in the average returns of all the Islamic and conventional portfolios, while the difference in variances is not significant for almost all sectors, except two: Financials and Utilities. For these latter, the variance is higher with Islamic portfolios than with the conventional ones. This result thus suggests that investors in these two sectors prefer conventional portfolios if they are risk-averse and Islamic portfolios if they are risk-seeking. From Table 6 (PF3, minimal-variance portfolios,), we find that the results are the same as those for PF2 in Table 5. This means that there is no significant difference in the mean returns, while there are significant differences for the variances in the Financial and Utilities sectors. The results from Table 7 (PF4, CCC-GARCH portfolios) are also consistent with those from Tables 5 and 6.

Thus, the MR analysis, on the comparison between Islamic and conventional portfolios, leads to the following conclusions. First, we find that for the portfolios without gold (PF1), risk-averse investors prefer conventional portfolios (with the same return and smaller risk), while risk-seeking investors prefer the corresponding Islamic portfolios (with the same return and higher risk) in five sectors (50%). However, in the other five sectors, both risk-averse and risk-seeking investors are indifferent between the Islamic and conventional portfolios over the entire period. Second, the inclusion of gold in these portfolios following different strategies (PF2, PF3, and PF4) results in more indifference between the Islamic and conventional portfolios. There are only two sectors (Financials and Utilities) in which risk-averse investors would be better off choosing the conventional portfolios, while risk-seeking investors would be better off choosing Islamic. Regarding these sectorial differences, Charles and Darné (2015) find that the Islamic indexes outperform their conventional counterparts in the Basic Materials, Consumer Goods and Services, Health Care, Industrials, Technologies and Telecommunications sectors. They argue that Islamic screens lead to a higher concentration in some sectors, especially Basic Materials, Industrials and Technology in most DJ Islamic indexes. In addition, within the four diversification strategies understudy, the equal-weighted portfolio (PF2) gives the highest rate of return confirming the finding of De Miguel et al.

(2009). However, the portfolios that follow the Markowitz approach (PF3) provide the lowest variance.

The following section presents the results on the second comparison, between portfolios with gold (PF2, PF3, PF4) and those without gold (PF1), using the MR method.

5.2.2. Mean-risk comparisons between portfolios with and without gold

The results on conventional portfolios are presented in Panel A and those on Islamic portfolios in Panel B of Table 8.

[Insert Table 8 here]

We first focus on the conventional portfolios, with and without gold. From Panel A, the results of the t-test show that there is no significant difference in the average returns between the portfolios with gold (PF2, PF3, PF4) and the portfolio without gold (PF1). However, the results from the F-test show that the variances of returns of portfolios with gold are significantly lower than that of portfolios without gold in all sectors. This means that in the conventional portfolios, gold helps to reduce the volatility of returns. This result is consistent with the findings in Hoang et al. (2015a) and Hoang et al. (2015b) for French and Chinese conventional portfolios, respectively. The results for Islamic portfolios (Table 8, Panel B) remain the same, meaning that risk-averse investors prefer Islamic portfolios with gold and risk-seeking investors prefer Islamic portfolios without gold. The conclusion drawn from the mean-MVaR is similar to that from the MV analysis.

Overall, the MR analysis on the comparison between portfolios with and without gold leads to the following conclusions. Regardless of the diversification strategies (PF2, PF3, and PF4), portfolios with gold dominate portfolios without gold for risk-averters, while it is the reverse for risk-seekers. This conclusion holds for both the Islamic and the conventional portfolios and for any sector. This implies that following the MR analysis, the impact of gold remains the same either for conventional or Islamic portfolios. This evidence supports the argument by Sadorsy (2014) in distinguishing between conventional stocks and socially responsible stocks. The results regarding risk-averse investors are consistent with those of previous studies, such as Hoang et al. (2015a), Kumar (2014), Baur and Löffler (2015), and Choudhry et al. (2015), among others. As for risk-seeking investors, very few studies have investigated this kind of investors, so it is hard to establish a consistent comparison with previous studies. We also notice that the decrease of the variance is higher for PF3 (minimal-variance portfolios) and PF4 (CCC-GARCH portfolios) than for PF2 (50% stocks and 50% gold). This suggests that the strategies minimizing the variance of returns are more efficient

than the naïve strategy (equal-weighted portfolios, PF2). This finding is consistent with the findings from Hoang et al. (2015b) in the Chinese context.

5.3. Stochastic dominance comparisons

We first compare between Islamic and conventional portfolios (section 5.3.1) and follow by the comparison between portfolios with gold and those without gold (section 5.3.2).

5.3.1. SD comparisons between Islamic and conventional portfolios

Table 9 shows the comparison between Islamic and conventional portfolios within each strategy (PF1 to PF4). There is no first-order stochastic dominance relationship between the Islamic and conventional portfolios for any diversification strategy and for any sector. This implies that there is no arbitrage opportunity for Islamic and conventional portfolios. This supposes that Islamic and conventional markets are efficient (see Section 4). In addition, similar to the results obtained from the MR analysis, there is no second- or third-order SD relationship in some sectors while it is the case in some other sectors. However, different from the MR analysis, there are more dominance relationships when applying the SD approach, as detailed in the following paragraphs.

[Insert Table 9 here]

In the first portfolio (PF1, 100% stocks), the conventional portfolios second- and thirdorder stochastically dominate their Islamic counterparts for risk-averse investors. As for riskseeking investors, the preference is reversed in most sectors (seven over ten), except the Consumer Goods, Health Care and Telecommunications sectors for which there is no SD. This result can be explained by the fact that Islamic stocks are more volatile than their conventional counterparts (see Table 1). That is why risk-averse investors prefer conventional portfolios to Islamic ones, while the reverse is true for risk-seeking investors.

For the second portfolio (PF2, 50% stocks+50% gold), there is the same number of sectors for which there is no SD relationship and 3 more sectors for which there is no RSD relationship, compared to PF1. This suggests that including 50% gold in the portfolio may significantly change the relationship between Islamic and conventional stocks. The sectors for which there is no SD include Financials, Health Care and Industrials. For most of the other sectors, the principal result is: risk-averters prefer conventional portfolios, while risk-seekers prefer Islamic portfolios. However, the preferences are reversed in the Consumer Goods sector such that risk-averters prefer Islamic portfolios, while risk-seekers prefer conventional portfolios. The Consumer Services sector is different because there is no SD relationship for risk-seekers but risk-averters prefer Islamic portfolios. The same mixed result is also found

for the Telecommunications sector, for which there is no SD for risk-seekers, while riskaverters prefer conventional portfolios. Overall, the results on PF2 lead to the conclusion that including gold can significantly change the relationship between the Islamic and the conventional portfolios. In most sectors, risk-averse investors prefer conventional portfolios, while risk-seeking investors prefer Islamic portfolios.

For PF3 and PF4 (minimal-variance and CCC-GARCH portfolios), the results are similar to those for PF2: There are more sectors in which there is no SD and/or RSD relationship than in PF1. But different from PF2, both PF3 and PF4 have more sectors in which there is no SD relationship when comparing with PF2, but there are fewer sectors in which there is no RSD relationship when comparing with PF1. Furthermore, for PF3 and PF4, risk-averse investors prefer conventional portfolios, while risk-seeking investors prefer the Islamic counterparts in four sectors out of ten (Basic Materials, Technology, Telecommunications and Utilities). In other sectors, including Consumer Services, Financials, Health Care and Oil & Gas, there is no SD relationship. For the two remaining sectors, Consumer Goods and Industrials, the results are a bit different: There is no SD for risk-averse investors, while for risk-seeking investors, the conventional portfolios dominate in the Consumer Goods sector and the dominance is reversed in the Industrials sector.

Overall, the SD comparison between Islamic and conventional portfolios leads to the conclusion that in most sectors, risk-averse investors prefer conventional portfolios, while risk-seeking investors prefer Islamic portfolios. The next sub-section will focus on the SD comparison between portfolios with gold and those without gold (PF2, PF3 and PF4 to PF1).

5.3.2. SD comparisons between portfolios with and without gold

Table 10 shows that similar to the findings on the comparison between Islamic and conventional portfolios, there is no first-order SD. This suggests that there is no arbitrage opportunity between portfolios with and without gold and markets with and without gold are efficient (see Section 4). For both Islamic and conventional portfolios, those with gold stochastically dominate those without gold for risk-averse investors. On the contrary, risk-seekers prefer portfolios without gold to those with gold in almost all sectors. This result is true for all strategies with gold (PF2, PF3 and PF4) and for nearly all sectors. This thus confirms the theory that risk-averse investors prefer more diversified portfolios to reduce their risk, while risk-seeking investors prefer less diversified portfolios, which provide higher risks. This result also shows that it is important to distinguish between risk-averse and risk-seeking investors when studying gold investments.

[Insert Table 10 here]

Overall, the SD analyses on the above-mentioned double comparison lead to the following conclusions: First, there is no arbitrage opportunity. Second, the efficient market hypothesis may hold in these markets. Third, as with the findings from the MR analysis, risk-averters prefer investing in portfolios with gold, while risk-seekers prefer portfolios without gold for all strategies, for both Islamic and conventional portfolios, and for nearly all sectors. Fourth, in most sectors, the conclusion drawn from the SD analysis is consistent with the MR analysis: risk-averse investors prefer conventional portfolios, while risk-seeking investors prefer the corresponding Islamic portfolios. Fifth, the diversification strategies with gold can significantly change the relationships between Islamic and conventional portfolios. Sixth, the sectorial effect is significant and in most sectors, risk-averse investors prefer conventional portfolios. In other sectors, the preference is reversed or there is no SD relationship. Seventh, in most cases, adopting the strategy proposed by Markowitz (1952) or Kroner and Ng (1998) does not change the preference of investors. Eighth, there are differences between the naïve strategy (equal-weighted portfolios or PF2) and risk-minimizing strategies (PF3 and PF4).

The above conclusions are for the entire period (2002-2014). However, do these results remain true in different periods? It is the robustness check that we perform in the next section.

6. Robustness check: Are results time-dependent?

As in previous sections, we also conduct a double comparison (between Islamic and conventional portfolios and between portfolios with and without gold) in three different subperiods defined in Section 3. We recall that the first sub-period (bull market for all markets) is from December 31, 2002, to June 7, 2008; the second one (bull market for gold and bear market for stocks) is from June 8, 2008, to September 6, 2011; and the third one (bear market for gold and bull market for stocks) is from September 7, 2011, to December 31, 2014.

6.1. Comparisons between Islamic and conventional portfolios in sub-periods

6.1.1. Mean-risk comparisons¹⁴

Overall, the results of the MR analysis in each sub-period are similar to those for the entire period. For example, the average returns are not significantly different, between the conventional and Islamic portfolios for PF1, in all the three sub-periods. The variances of

¹⁴ The tables reporting the MR analysis for the double comparison (between Islamic and conventional portfolios, and between portfolios with gold and without gold, next sub-section) in the three sub-periods contain many panels. Thus, we only report the results and skip exhibiting the tables in the Appendix file. They are available upon request. The tables for SD comparisons are still reported in the Appendix file.

Islamic portfolios are either significantly higher than those of conventional portfolios or not significantly different for PF1 in all three sub-periods. Thus, the sub-period analysis confirms our first conclusion from the MR analysis for the entire period: Risk-averse investors prefer conventional portfolios, while risk-seeking investors prefer the corresponding Islamic portfolios in some sectors, while there is no preference in other sectors. As for portfolios with gold (PF2, PF3, PF4), the results are similar to those for PF1. However, it remains that including gold changes significantly the relationship between Islamic portfolios and conventional ones.

6.1.2. Stochastic dominance comparisons

Table 11 shows that in the 1st sub-period for PF1 (100% stocks), risk-averse investors prefer conventional portfolios; while risk-seeking investors prefer Islamic portfolios for almost all sectors (except the Consumer Goods sector). The situation changes in the 2nd and 3rd sub-periods for different sectors: either the preferences of risk-averse and risk-seeking investors change or there is no longer a SD relationship between the Islamic and conventional portfolios. We illustrate this by looking at the Health Care sector for which risk-averse investors prefer conventional portfolios, while risk-seeking investors prefer Islamic portfolios in the 1st sub-period. However, in the 2nd sub-period, the preference is reversed. Furthermore, in the 3rd sub-period, risk-averse investors prefer conventional portfolios is time-dependent.

[Insert Table 11 here]

6.2. Comparisons between portfolios with gold and without gold in sub-periods

In this sub-section, we will analyze the impact of time on the comparison between portfolios with gold (PF2, PF3 and PF4) and those without gold (PF1) in the three sub-periods. We first apply the MR approach and follow with the SD approach.

6.2.1. Mean-risk comparisons

First, the results of the t-test show that the average returns are not significantly different between portfolios with gold and those without gold in each sub-period in all cases. This is thus similar to the results of the whole period. Second, the results of the F-test show that the variances of returns for portfolios without gold are significantly greater than those of portfolios with gold (for all strategies, all sub-periods and both Islamic and conventional portfolios). Thus, similar to the whole period, the inclusion of gold helps to reduce the volatility of returns of portfolios. However, there are exceptions in some sectors, such as Consumer Goods, Consumer Services, Financials and Utilities. For example, in sub-period 1, for Islamic portfolios in the Utilities sector, there is no significant difference in the variances of returns between PF1 and PF2, PF3, and PF4. These exceptions can change in different sub-periods and in different sectors. Thus, we conclude that the time period can have an impact on some sectors. As for the MVaR, the finding is the same as that for the variance. These results are similar for conventional portfolios and Islamic portfolios. This suggests that following the MR approach, the impact of gold on Islamic and conventional stocks is not significantly different. In most cases, risk-averse investors prefer portfolios with gold, while risk-seeking investors prefer portfolios without gold. This result is similar to that for the whole period for most cases. Finally, this result leads us to suggest that based on the MR approach, the impact of the time period on the comparison between portfolios with and without gold is not significant in most cases, but it can be significant for certain sectors, especially for the Financial and Utilities sectors.

6.2.2. Stochastic dominance comparisons

Last, we present the results obtained by using the stochastic dominance approach in Table 12 to compare portfolios with gold (PF2, PF3, PF4) to those without gold (PF1) (for both Islamic and conventional portfolios in each sector and in each sub-period). The table reveals that the results for sub-periods 1 and 2 are very similar to those for the whole period. This means that in the first two sub-periods, for most cases, risk-averse investors prefer portfolios with gold, while risk-seeking investors prefer portfolios without gold. On the other hand, the results of the third sub-period are very different for risk-averse investors, since, in most cases, there is no SD relationship between portfolios with gold and those without gold. For risk-seeking investors, the results remain the same as those for the whole period in almost all cases. This may suggest that the tendency of gold prices to decrease from 2011 to 2014 may have changed the behavior of risk-averse investors but not that of risk-seeking investors. Finally, these results show that following the SD approach, the results are time-varying. These findings on sub-periods also suggest that gold can be considered as a safe haven asset for riskaverters in the first two sub-periods when the price of gold price was on an upward trend. However, it is not the case in the third sub-period when gold was on a downward trend. In the latter case, both risk-averse and risk-seeking investors prefer to invest in portfolios without gold to those with gold.

[Insert Table 12 here]

7. Conclusions

Through this research, we have found that it is important to use the stochastic dominance (SD) method to compare between Islamic and conventional portfolios. Indeed, the mean-risk (MR) approach (mean-variance and mean-MVaR) leads to the conclusion that the difference between them is not significant (except for the Financials and Utilities sectors). However, following the SD approach, we find that there is a SD relationship between Islamic and conventional portfolios in most sectors. Regarding the impact of gold, the MR results show that including gold does not change the relationship between Islamic and conventional portfolios, while the SD approach shows the contrary. These findings thus suggest that it is important to use a robust method, such as stochastic dominance, to compare Islamic to conventional stocks and to compare portfolios with gold to those without gold.

Following the MR results, risk-averse investors prefer conventional portfolios, while riskseeking investors prefer the corresponding Islamic portfolios in only two sectors (Financials and Utilities). However, following the SD results, this is the case in most sectors. This result suggests that Islamic portfolios represent a higher risk and thus correspond more to investors being risk-seeking. As for the benefit of gold in the diversification of portfolios, the comparison of PF2, PF3, PF4 (with gold) to PF1 (without gold) shows that, in most cases, risk-averse investors prefer portfolios with gold, while risk-seeking investors prefer portfolios without gold. This is true for both Islamic and conventional portfolios. This finding thus confirms the theory that risk-averse investors prefer more diversified portfolios in order to reduce the risk. Our robustness check shows that these results are time-dependent and gold tends to be more profitable in down-trending periods for stocks. This thus confirms the role of gold as a safe haven highlighted in numerous past studies. Furthermore, there are significant sectorial differences, especially with the Financials and Utilities sectors when comparing Islamic to conventional stocks. Finally, as expected, the equal-weighted portfolio (50%) stocks+50% gold) has different characteristics compared to the minimal-variance portfolio (Markowitz, 1952) and CCC-GARCH portfolio (Kroner and Ng, 1998). These results reveal interesting managerial implications in terms of portfolio diversification regarding investors' beliefs and risk-aversion levels.

References

Abdelsalam, O., Duygun, M., Matallin-Saez, J.C., Tortosa-Ausina, E., 2014. Do ethics imply persistence? The case of Islamic and socially responsible funds. *Journal of Banking & Finance* 40,182-194

- Abdullah, A.M., Saiti, B., Masih, M., 2016. The impact of crude oil price on Islamic stock indices of South East Asian countries: Evidence from MGARCH-DCC and Wavelet approaches. *Borsa Istanbul Review*, In Press.
- Abedifar, P., Molyneux, P., Tarazi, A., 2013. Risk in Islamic banking. *Review of Finance* 17(6), 2035-2096.
- Abedifar, P., Hassan, I. Tarazi, A., 2016. Finance-growth nexus and dual banking systems: Relative importance of Islamic banks. *Journal of Economic Behavior & Organization*, In Press.
- Abul-Hassan, A.A., Krishna, P.D., 2005. Impact of ethical screening on investment performance: The case of Dow Jones Islamic index. *Islamic Economic Studies* 12-13, 69-97.
- Ahmad, Z., Ibrahim, H., 2002. A study of performance of the KLSE Syariah index. *Malaysian Management Journal* 6(1&2), 25-34.
- Al-Khazali, O., Lean, H.H., Samet, A., 2014. Do Islamic stock indexes outperform conventional stock indexes? A stochastic dominance approach. *Pacific-Basin Finance Journal* 28, 29-46.
- Arouri, M.E.H., Jouini, J., Nguyen, D.K., 2011. Volatility spillover oil prices and stock sector returns: implications for portfolio management. *Journal of International Money and Finance* 30, 1387-1405.
- Arouri, M.E.H., Ben Ameur, H., Jawadi, F., Jawadi, N., Louhichi, W., 2013. Are Islamic finance innovations enough for investors to escape from a financial downturn? Further evidence from portfolio simulations. *Applied Economics* 45(24), 3412-3420.
- Bai, Z.D., Li, H., McAleer, M., Wong, W.K., 2015. Stochastic dominance statistics for risk averters and risk seekers: an analysis of stock preferences for USA and China. *Quantitative Finance* 15(5), 889-900.
- Bai, Z.D., Liu, H.X., Wong, W.K., 2009. Enhancement of the applicability of Markowitz's portfolio optimization by utilizing random matrix theory. *Mathematical Finance* 19(4), 639-667.
- Barrett, G.F., Donald, S.G., 2003. Consistent tests for stochastic dominance. Econometrica 71, 71-104.
- Barunik, J., Kocenda, E., Vacha, L., Gold, oil and stocks: dynamic correlations. *International Review* of Economics and Finance 42, 186-201.
- Basher, S.A., Sadorsy, P., 2016. Hedging emerging market stock prices with oil, gold VIX, and bonds: A comparison between DCC, ADCC and GO-GARCH. *Energy Economics* 54, 235-247.
- Baur, D.G., Löffler, G., 2015. Predicting the equity premium with the demand for gold coins and bars. *Finance Research Letters* 13, 172-178.
- Baur, D.G., 2011. Explanatory mining for gold: Contrasting evidence from simple and multiple regressions. *Resources Policy* 36, 265-275.
- Baur, D.G., Lucey, B.M., 2010. Is gold a hedge or a safe haven? An analysis of stocks, bonds and gold. *The Financial Review* 45, 217-2209.
- Baur, D.G., McDermott, T.K., 2010. Is gold a safe haven? International evidence. *Journal of Banking & Finance* 34, 1886-1898.
- Beck, T., Demirgüç-Kunt, A., Merrouche, O., 2013. Islamic vs. conventional banking: Business model, efficiency and stability. *Journal of Banking & Finance* 40, 182-194.
- Beckmann, J., Berger, T., Czudaj, R., 2015. Does gold act as a hedge or a safe haven for stocks? A smooth transition approach. *Economic Modelling* 48, 16-24.
- Bishop, J.A., Formly, J.P., Thistle, P.D., 1992. Convergence of the South and non-South income distributions. *American Economic Review* 82, 262–272. 427
- Blose, L.E., 1996. Gold price risk and the returns on gold mutual funds. *Journal of Economics and Business* 48, 499-513.
- Blose, L.E., Shieh, J.C.P., 1995. The impact of gold price on the value of gold mining stock. *Review of Financial Economics* 4(2), 125-139.
- Bollerslev, T., 1990. Modelling the coherence in short-run nominal exchange rates: A multivariate generalized ARCH model. *Review of Economics and Statistics* 72, 498-505.
- Bredin, D., Conlon, T., Potì, V., 2015. Does gold glitter in the long-run? Gold as a hedge and safe haven across time and investment horizon. *International Review of Financial Analysis* 41, 320-328.
- Caporin, M., Jannin, G., Lisi, F., Maillet, B., 2014. A survey on the four families of performance measures. *Journal of Economic Survey* 28(5), 917-942.
- Chan, R.H., Clark, E., Wong, W.K., 2012a. On the third order stochastic dominance for risk-averse and risk-seeking investors. MPRA Paper 42676, University Library of Munich, Germany

- Chan, C.Y., de Peretti, C., Qiao, Z., Wong, W.K., 2012b. Empirical test of the efficiency of the UK covered warrants market: stochastic dominance and likelihood ratio test approach. *Journal of Empirical Finance* 19(1), 162-174.
- Charles, A., Darné, O., 2015. Are the Islamic indexes size or sector oriented? Evidence from Dow Jones Islamic indexes. *Economics Bulletin* 35(3), 1897-1905.
- Chkili, W., 2016. Dynamic correlations and hedging effectiveness between gold and stock markets: evidence for BRICS countries. *Research in International Business and Finance* 38, 22-34.
- Choudhry, T., Hassan, S.S., Shabi, S., 2015. Relationship between gold and stock markets during the global financial crisis: Evidence from nonlinear causality tests. *International Review of Financial Analysis* 41, 247-256.
- Chua, J.H., Sick, G., Woodward, R.S., 1990. Diversifying with gold stocks. *Financial Analysts Journal* 46(4), 76-79.
- Cihak, M., Hesse, H. 2010. Islamic banks and financial stability: An empirical analysis. *Journal of Financial Services Research* 38, 95-113
- Ciner, C., Gurdgiev, C., Lucey, B.M., 2013. Hedges and safe havens: an examination of stocks, bonds, gold, oil and exchange rates. *International Review of Financial Analysis*, 29, 202-211.
- Creti, A., Joëts, M., Mignon, V., 2013. On the links between stock and commodity markets' volatility. *Energy Economics* 37, 16-28.
- Davidson, R., Duclos, J.Y., 2000. Statistical inference for stochastic dominance and for the measurement of poverty and inequality. *Econometrica* 68, 1435-1464.
- Davidson, S., Faff, R., Hillier, D., 2003. Gold exposures in international asset pricing. *Journal of International Financial Markets, Institutions and Money* 13, 271-289.
- Derigs, U., Marzban, S., 2009. New strategies and a new paradigm for Shariah-compliant portfolio optimization. *Journal of Banking & Finance*, 33, 1166–1176.
- Dewandaru, G., Ismath Bacha, O., Masih, A.M.M, Masih, R., 2015. Risk-return characteristics of Islamic equity indices: Multi-timescales analysis. *Journal of Multinational Financial Management* 29, 115-138
- De Miguel, V., Garlappi, L., Uppal, R., 2009. Optimal versus naive diversification: How inefficient is the 1/N portfolio strategy? *Review of Financial Studies* 22(5), 1915-1953.
- Dow Jones. 2015. Dow Jones Islamic market indices methodology. <u>https://www.djindexes.com/mdsidx/downloads/meth_info/Dow_Jones_Islamic_Market_Indices_M</u> <u>ethodology.pdf</u> (accessed February 2016).
- El gari, M.A., 1993. Towards an Islamic stock market. Islamic Economic Studies, 1(1), 1-20.
- El khamlichi A., Sarkar, H.K., Arouri, M., Teulon, F., 2014a. Are Islamic equity indices more efficient than their conventional counterparts? Evidence from major global index families. *Journal of Applied Business Research* 30 (4), 1137-1150.
- El khamlichi, A., Sannajust, A., Sarkar, H.K., 2014b. Islamic equity indices: Insight and comparison with conventional counterparts. *Bankers, Markets and Investors* 130, 69-80.
- Engle, R.F., Kroner, K.F., 1995. Multivariate simultaneous generalized ARCH. *Economic Theory* 11, 122-150.
- Falk, H., Levy, H., 1989. Market reaction to quarterly earnings' announcements: A stochastic dominance based test of market efficiency. *Management Science* 35, 425-446.
- Flavin, T.J., Morley, C.E., Panopoulou, E., 2014. Identifying safe havens assets for equity investors through an analysis of the stability of shock transmission. *Journal of International Financial Markets, Institutions and Money* 33, 137-154.
- Fong, W.M., Lean, H.H., Wong, W.K., 2008. Stochastic dominance and behavior towards risk: The market for Internet stocks. *Journal of Economic Behavior and Organization* 68(1), 194-208.
- Gasbarro, D., Wong, W.K., Zumwalt, J.K., 2007. Stochastic dominance analysis of iShares. *European Journal of Finance* 13, 89–101.
- Ghazali, M-F, Lean, H.H., Bahari, Z., 2015. Sharia compliant gold investment in Malaysia: Hedge or safe haven? *Pacific-Basin Finance Journal* 34, 192-204.
- Ghoul, W., Karam, P., 2007. MRI and SRI mutual funds: A comparison of Christian, Islamic (morally responsible investing), and socially responsible investing (SRI) mutual funds. *The Journal of Investing* 16 (2), 96-102.

- Girard, E., Hassan, M. K., 2008. Is there a cost to faith-based investing: evidence from FTSE Islamic indices? *The Journal of Investing* 17 (4), 112-121.
- Gregoriou, G.N., Gueyie, J.P., 2003. Risk-adjusted performance of funds of hedge funds using a modified Sharpe ratio. *Journal of Wealth Management* 6(3), 77-83.
- Guo, X., Post, T., Wong, W.K., Zhu, L.X., 2014. Moment conditions for almost stochastic dominance. *Economics Letters*, 124(2), 163–167.
- Guo, X., Wong, W.K., 2016. Multivariate stochastic dominance for risk averters and risk seekers. *RAIRO Operations Research*, Forthcoming.
- Guo, X., Wong, W.K., Zhu, L.X., 2016. Almost stochastic dominance for risk averters and risk seekers. *Finance Research Letters*, Forthcoming.
- Guyot, A., 2011. Efficiency and dynamics of Islamic investment: evidence of geopolitical effects on Dow Jones Islamic market indexes. *Emerging Markets Finance and Trade* 47 (6), 24–45.
- Haddad, M., Homaifar, G., Elfakhani, S., Ahmedov, H., 2009. Intertemporal test of beta stationarity performance of Islamic sector structured mutual funds. *Journal for International Business and Entrepreneurship Development* 4 (4), 275-285.
- Hammoudeh, S., Santos, P.A., Al-Hassan, A., 2013. Downside risk management and VaR-based optimal portfolios for precious metals, oil and stocks. *The North American Journal of Economics and Finance* 25, 318-334.
- Hayat, R., Kraeussl, R., 2011. Risk and return characteristics of Islamic equity funds. *Emerging Markets Review* 12 (2), 189-203.
- Hoang, T., Hooi, L.L., Wong, W.K., 2015a. Is gold good for portfolio diversification? A stochastic dominance analysis of the Paris stock exchange. *International Review of Financial Analysis* 42, 98-108.
- Hoang, T., Wong, W.K., Zhu, Z.Z., 2015b. Is gold different for risk-averse and risk-seeking investors? An empirical analysis of the Shanghai Gold Exchange. *Economic Modelling* 50, 200-211.
- Hoang, T., Lahiani, A., Heller, D., 2016. Is gold a hedge against inflation? New evidence from a nonlinear ARDL approach. *Economic Modelling* 54, 54-66.
- Hood, M., Malik, F., 2013. Is gold the best hedge and a safe haven under changing stock market volatility? *Review of Financial Economics* 22, 47-52.
- Hussein, K., Omran, M., 2005. Ethical investment revisited: evidence from Dow Jones Islamic indexes. *The Journal of Investing* 14 (3), 105-124.
- Iqbal, M., 2002. *Islamic banking and finance: current developments in theory and practice*. Islamic Foundation, Leicester, UK.
- Ismail, A. G., Shakrani, M. S., 2003. The conditional CAPM and cross-sectional evidence of return and Beta for Islamic unit trusts in Malaysia. *Journal of Economics and Management* 11 (1), 1-31.
- Jaffe, J.F., 1989. Gold and gold stocks as investments for institutional portfolios. *Financial Analysts Journal* 49(2), 53-59.
- Jain, A., Biswal, P.C., 2016. Dynamic linkages among oil price, gold price, exchange rate, and stock market in India. *Resources Policy* 49, 179-185.
- Jarrow, R., 1986. The relationship between arbitrage and first order stochastic dominance. *The Journal* of *Finance* 41, 915-921.
- Jawadi, F., Jawadi, N., Cheffou, A.I., 2015. Are Islamic stock markets efficient? A time-series analysis. *Applied Economics* 47(1), 1686-1697.
- Jorion, P., 2000. Value at Risk: The new benchmark for managing financial risk. New York, NY: McGraw-Hill.
- Kaabia, O., Abid, I., Mkaouar, F., 2016. The dark side of the black gold shock onto Europe: One stock's joy is another stock sorrow. *Economic Modelling*, In Press.
- Kenourgios, D., Naifar, N., Dimitriou, D., 2016. Islamic financial markets and global crises: Contagion or decoupling? *Economic Modelling* 57, 36-46.
- Kroner, K.F., Ng, V.K., 1998. Modelling asymmetric movements of asset prices. *Review of Financial Studies* 11, 844-871.
- Kumar, D., 2014. Return and volatility transmission between gold and stock sectors: Application for portfolio management and hedging effectiveness. *IIMB Management Review* 26, 5-16.

- Lean, H.H., Wong, W.K., Zhang, X.B., 2008. The size and power of some stochastic dominance tests: A Monte Carlo study for correlated heteroskedastic distributions. *Mathematics and Computer in Simulation* 79, 30-48.
- Lee, G.H.Y., 2011. Gold dinar for the Islamic countries? Economic Modelling 28, 1573-1586.
- Leshno, M., Levy, H., 2002. Preferred by "all" and preferred by "most" decision makers: Almost stochastic dominance. *Management Science* 48(8), 1074-1085.
- Leung, P.L., Ng, H.Y., Wong, W.K., 2012. An improved estimation to make Markowitz's portfolio optimization theory users friendly and estimation accurate with application on the US stock market investment. *European Journal of Operational Research* 222, 85–95.
- Levy, H., 2015. *Stochastic dominance: Investment decision making under uncertainty*. Third edition, Berlin: Springer.
- Li, C.K., Wong, W.K., 1999. Extension of stochastic dominance theory to random variables. *RAIRO Operations Research* 33(4), 509-524.
- Lucey, B., Tully, E., Poti, V., 2006. International portfolio formation, skewness and the role of gold. *Frontiers in Finance and Economics* 3(1), 49-67.
- Ma, C., Wong, W.K., 2010. Stochastic dominance and risk measure: A decision-theoretic foundation for VaR and C-VaR. *European Journal of Operational Research* 207(2), 927–935.
- Malliaris, A.G., Malliaris, M., 2015. What drives gold returns? A decision tree analysis. *Finance Research Letters* 13, 45-53.
- Mansor, F., Bhatti, M.I., 2011. *Islamic mutual funds' performance for emerging markets during bullish and bearish: The case of Malaysia*. 2nd International Conference on Business and Economic Research, 14-16 March, Langkawi, Kedah, Malaysia.
- Markowitz, H.M., 1952. Portfolio selection. The Journal of Finance 7, 77-91.
- McDonald, J.G., Solnik, B.H., 1977. Valuation and strategy for gold stocks. *The Journal of Portfolio Management* 3, 29-33.
- Mensi, W., Hammoudeh, S., Reboredo, J.C., Nguyen, D.K., 2015. Are Sharia stocks, gold and U.S. Treasury hedges and/or safe havens for the oil-based GCC markets? *Emerging Markets Review* 24, 101-121.
- Michis, A.A., 2014. Investing in gold: Individual asset risk in the long run. *Finance Research Letters* 11, 369-374.
- MIFC 2015. *Islamic Funds Industry: 2015 Review & Outlook*. Available at: http://www.mifc.com/index.php?ch=28&pg=72&ac=159&bb=uploadpdf [accessed January 2016]
- Mohammad, N., Ashraf, D., 2015. The market timing ability and return performance of Islamic equities: An empirical study. *Pacific-Basin Finance Journal* 34, 169–183
- Mollah, S. and Zaman, M. 2015. Shari'Ah supervision, corporate governance and performance. *Journal of Banking & Finance* 59, 418-435.
- Morales, L., Andreosso-O'Callaghan, B., 2011. Comparative analysis on the effects of the Asian and global financial crises on precious metal markets. *Research in International Business and Finance* 25, 203-227.
- Muhammad, N. M. N., Mokhtar M., 2008. Islamic equity mutual fund performance in Malaysia: risk and return analysis. *Proceeding of The Midwest Finance Association (MFA) Conference*, 211-243, February 27-March 1, San Antonio, USA.
- Nagayev, R., Disli, M., Inghelbrecht, K., Ng, A., 2016. On the dynamic links between commodities and Islamic equity? *Energy Economics* 58, 125-140.
- Narayan, P.K., Narayan, S., Sharma, S.S., 2013. An analysis of commodity markets: What gain for investors? *Journal of Banking and Finance* 37, 3878-3889.
- Narayan, P.K., Ali Ahmed, H., Narayan, S., 2015. Do momentum-based trading strategies work in the commodity futures markets? *Journal of Futures Markets* 35, 868-891.
- Narayan, P.K., Bannigidadmath, D., 2015. Does financial news predict stock returns? New evidence from Islamic and non-Islamic stocks. *Pacific-Basin Finance Journal*, In Press.
- Naughton, S., Naughton, T., 2000. Religion, ethics and stock trading: the case of an Islamic equities markets. *Journal of Business Ethics* 23 (2), 145-159.
- Nguyen, C., Bhatti, M.I., Komornikova, M., Komornik, J., 2016. Gold price and stock markets nexus under mixed-copulas. *Economic Modelling* 58, 283-292.

- Obaidullah, M., 2001. Ethics and efficiency in Islamic stock markets. *Islamic Journal of Financial Services* 3 (2), 145-159.
- O'Connor, F.A., Lucey, B.M., Batten, J.A., Baur, D.G., 2015. The financial economics of gold A survey. *International Review of Financial Analysis* 41, 186-205.
- Ogryczak, W., Ruszczynski, A., 2002. Dual stochastic dominance and related mean-risk models. *SIAM Journal on Optimization* 13(1), 60-78.
- Qiao, Z., Wong, W.K., Fung, J.K.W., 2013. Stochastic dominance relationships between stock and stock index futures markets: International evidence. *Economic Modelling* 33, 552–559.
- Raza, N., Shahzad, S.J.H., Tiwari, A.K., Shahbaz, M., 2016. Asymmetric impact of gold, oil prices and their volatilities on stock prices of emerging markets. *Resources Policy* 49, 290-301.
- Rockafellar, R.T., Uryasev, S.P., 2000. Optimization of conditional Value-at-Risk. *Journal of Risk* 2, 21-42.
- Sadorsy, P., 2014. Modeling volatility and conditional correlation between socially responsible investments, gold and oil. *Economic Modelling* 38, 609-618.
- Sensoy A., Aras G., Hacihasanoglu E., 2015, Predictability dynamics of Islamic and conventional equity markets, *The North American Journal of Economics and Finance* 31, 222-248.
- Shamsuddin, A., 2014. Are Dow Jones Islamic equity indices exposed to interest rate risk? *Economic Modelling* 39, 273-281.
- Sherman, E.J., 1982. Gold: A conservative, prudent diversifier. *The Journal of Portfolio Management* 8(3), 21-27.
- Soucek, M., 2013. Crude oil, equity and gold futures open interest co-movements. *Energy Economics* 40, 306-315.
- Sriboonchitta, S., Wong, W.K., Dhompongsa, S., Nguyen, H.T., 2009. *Stochastic Dominance and Applications to Finance, Risk and Economics*. Chapman and Hall/CRC, Taylor and Francis Group.
- Walkshäusl, C., Lobe, S., 2012. Islamic investing. *Review of Financial Economics* 21(2), 53-62
- Wong, W.K., 2007. Stochastic dominance and mean-variance measures of profit and loss for business planning and investment. *European Journal of Operational Research* 182, 829-843.
- Wong, W.K., Chan, R., 2008. Markowitz and prospect stochastic dominance. *Annals of Finance* 4(1), 105-129.
- Wong, W.K., Phoon, K.F., Lean, H.H. 2008. Stochastic dominance analysis of Asian hedge funds. *Pacific-Basin Finance Journal* 16(3), 204–223.
- Zhang, Y., 2015. The securitization of gold and its potential impact on gold stocks. *Journal of Banking & Finance* 58, 309-326.